Author's response to reviews

Title: The Skiers Knee without Swelling or Instability - A Difficult Diagnosis.

Authors:

Mark E O'Donnell (modonnell904@hotmail.com)
Stephen A Badger (stephen@badger.tc)
David Campbell (David.Campbell@bch.n-i.nhs.uk)
Willie Loan (wloan@doctors.org.uk)
Brendan Sinnott (brendan.sinnott@bch.n-i.nhs.uk)

Version: 5 Date: 23 March 2007

Author’s response to reviews: see over
Dear Sir / Madam,

Re : Submission of Manuscript Revision.

“The Skiers Knee without Swelling or Instability – A Difficult Diagnosis”.

I am writing to you regarding the amendments requested for the revision of the above manuscript. Please find enclosed a re-submission with this covering letter documenting all actions as requested by the reviewers.

Reviewer 1

Abstract:
The abstract has been enhanced to include more information on the actual clinical case (Abstract: Page 3).

Case Report:
The patient’s body mass and height have been included (Case Report: Page 4, paragraph 1, line 4).
Review 2:

There are no specific revisions documented. However, the conclusion has been amended to include some of the reviewer’s comments.

“The presence of persistence knee pain following a high-speed injury should alert the physician to consider further investigations even in the absence of obvious clinical signs or radiological findings. Treatment should be symptomatic with an initial period of immobilization recommended.” (Conclusion – Page 9, para 1).

I greatly appreciate your consideration of the above manuscript for publication and please do not hesitate to contact me if there are any further queries. I enclose the amended manuscript with a copy of this cover letter.

Yours Faithfully,

Mark O’Donnell DSEM MB(Dist) MFSEM(UK) MRCSEd.